Author's response to reviews

Title: Multi-centre parallel arm randomised controlled trial to assess the effectiveness and cost-effectiveness of a group-based cognitive behavioural approach to managing fatigue in people with multiple sclerosis.

Authors:

Peter W Thomas (peter.thomas@poole.nhs.uk)
Sarah Thomas (sarah.thomas@poole.nhs.uk)
Paula Kersten (p.kersten@soton.ac.uk)
Rosemary Jones (Rosemary.Jones@nbt.nhs.uk)
Alison Nock (alison.nock@poole.nhs.uk)
Vicky Slingsby (vicky.slingsby@poole.nhs.uk)
Colin Green (colin.green@pms.ac.uk)
Roger Baker (roger.baker@poole.nhs.uk)
Kathleen Galvin (kgalvin@bournemouth.ac.uk)
Charles Hillier (charles.hillier@poole.nhs.uk)

Version: 4 Date: 15 June 2010

Author's response to reviews: see over
Typographical changes made:

Pg. 12 “and” removed from sentence
“exercise, work, family and social life

Pg. 15 sentence order changed to
Participants who are currently ineligible (e.g. due to medication recently started, a planned holiday, having received fatigue advice etc.) but who may fulfil eligibility criteria at a later date will be held over to be rescreened (if another iteration is scheduled in the centre).

Pg. 23 Sentence rewritten in a different tense
It is possible that the group-based nature of the intervention will result in observations that are clustered. This could result in…..

Pg. 30 Authors contributions
Sentence rearranged to:
ST is the Trial Coordinator and had significant input into the trial design, protocol development/writing and grant application.

References
Reference 61: full-stop removed after journal name

Reference 67: Capital letter added to “Thousand Oaks”